Author’s response to reviews

Title: Scoping studies: Advancing the methodology

Authors:

Danielle Levac (levacde@mcmaster.ca)
Heather Colquhoun (colquhhl@mcmaster.ca)
Kelly O’Brien (kelly.obrien@utoronto.ca)

Version: 4 Date: 14 September 2010

Author’s response to reviews:

References: Please make sure references are cited in numerical order. We notice reference 11 is cited before reference 10 within the main text. Please make sure references continue to be cited in numerical order once this reference is cited correctly.

This was done.

References in Tables: Please make sure that such references are cited in numerical order within the table

This request was addressed in email correspondence and determined that it did not require changes.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Author list: please amend the capitalisation of your author list in the submissions system so that it matches that in the manuscript. The preferred style is ‘First name middle Initial Last name’ (e.g. Joe F Bloggs)

This was done.

Tables- please remove the visible vertical lines from your tables

This was done.

References: please provide the full citation details for reference 9(In Press). You can view the BMC reference guide at the following link http://www.biomedcentral.com/bmcgenet/ifora/#references

This request was addressed in email correspondence and determined that it did not require changes.